

Vol. 454

Pretoria, 3 April 2003

No. 24708







No.

Page Gazette No. No.

GENERAL NOTICE

Independent Communications Authority of South Africa

General Notice

984 Telecommunications Act (103/1996): Notice of intention to amend the telecommunications service licences issued 24708 to Sentech Limited.....

GENERAL NOTICE

NOTICE 984 OF 2003



INDEPENDENT COMMUNICATIONS AUTHORITY OF SOUTH AFRICA

NOTICE OF INTENTION TO AMEND THE TELECOMMUNICATIONS SERVICE LICENCES ISSUED TO SENTECH LIMITED IN TERMS OF SECTION 32C (6)

The Independent Communications Authority of South Africa ("the Authority") hereby

- gives notice of applications received from Sentech Limited(Sentech) to amend their carrier of carriers and multimedia service licences in terms of section 48(1)(e) read with sections 34(3), 34(4), 34(5) and 35 of the Telecommunications Act, No.103 of 1996, as amended (hereinafter referred to as "the Act"). Copy of the application annexed herein;
- (2) Sentech is a holder of licences to provide an international telecommunications gateway service enabling it to operate as a carrier of carriers (the carrier of carriers licence) in terms of section 32C(1)(a) and a multimedia services licence (multimedia service licence) in terms of section 32C(1)(b) of the Act. Both licences were issued to Sentech by the Authority on 7 May 2002 in terms of section 32C(6) of the Act and were duly published in Government Gazette No. 23405 under Notice 686 dated 8 May 2002. Copies of both licences are annexed herein

- (3) The Authority hereby invites written comments from interested parties within 30 days from date hereof with regard to the proposed amendments.
- (4) The Authority further gives notice that written representations will be made publicly available except where respondents request that their responses or parts thereof be kept confidential. Respondents are requested to separate any confidential material into a clearly marked annexure Marked confidential.
- (5) Persons submitting written representations are invited to indicate in the first page of their submission whether they would like to make oral representations, which shall be limited to 1 hour.
- (6) Interested persons are hereby invited to submit written representations on the proposed amendments by Sentech. These have to be forwarded by post, hand delivery or fax and also in electronic format (Microsoft Word 6.0, Adobe PDF) for the attention of:

Mr Phineas Moleele – Manager: Telecommunication Licensing ICASA Private Bag X10002, Sandton, 2046; Block C, ICASA, Pin Mill Farm, 164 Katherine Street, Sandton;

Telephone (011)3218228,

Facsimile (011) 321 8550;

e-mail: pmoleele@icasa.org.za

BEFORE THE INDEPENDENT COMMUNICATIONS AUTHORITY OF SOUTH AFRICA ("THE AUTHORITY")

In the ex parte matter of:

SENTECH LIMITED

Applicant

APPLICATION IN TERMS OF SECTION 48(1)(e) IN RESPECT OF SENTECH'S CARRIER OF CARRIERS SERVICE LICENCE

1 THE APPLICANT

- 1.1 The Applicant is Sentech Limited ("Sentech"), a company established pursuant to the Sentech Act 63 of 1996 and incorporated in accordance with the company laws of the Republic of South Africa and having its principal place of business and registered office at Octave Street, Radiokop, Honeydew, Gauteng Province, South Africa.
- 1.2 Sentech is the holder of a licence to provide an international telecommunication gateway service enabling it to operate as a carrier of carriers in terms of section 32C(1)(a) ("the carrier of carriers licence") to the Telecommunications Act 103 of 1996 ("the Act"). The carrier of carriers licence was issued to Sentech by the Independent Communications Authority of South Africa ("the Authority") in terms of section 32C(6) of the Act on 7 May 2002. The carrier of carriers licence was subsequently published under Notice 684 of 2002 in Government Gazette No. 23405, dated 8 May 2002.
- 1.3 Sentech is also the holder of a licence to provide multimedia services in terms of section 32C(1)(b) to the Act ("the multimedia services licence"). The multimedia services licence was issued to Sentech by the Authority on 7 May 2002 in terms of section 32C(6) of the Act and was also subsequently published under Notice 686 of 2002 in Government Gazette No. 23405 dated 8 May 2002.

2 THE APPLICATION

2.1 Sentech has brought this application in terms of section 48(1)(e) to the Act, as it requires section 3.1 to its carrier of carriers licence to be amended by the addition of the word "lease" and to read as follows:-

"The Licensee is authorised by this Licence to procure, own, lease, construct, maintain, operate and use an International Telecommunication Gateway Facility for the provision of a Carrier of Carriers service within and outside the Territory."

- 2.2 Throughout the public process in respect of the grant of the carrier of carrier licence, Sentech requested the Authority to include the word "lease" in clause 3.1. The primary reason for this request, was due to the fact that the hire of facilities and other equipment is a core activity which Sentech should be authorised to provide in respect of its international telecommunications gateway facility. In addition, Sentech should also be entitled to lease its international telecommunications gateway facility for the provision of a carrier of carrier service both within and outside the Republic.
- 2.3 The word "lease" is included in clause 3.1 to the multimedia services licence and there should be no reason for this omission in the carrier of carriers licence.
- 2.4 Sentech is not in law prohibited from leasing its international gateway facility for the provision of a carrier of carriers service and there is no statutory justification for the inclusion of this limitation in the licence.
- 2.5 Should section 3.1 to the carrier of carriers licence not be amended to include the word "lease", Sentech will be limited in a manner not intended in terms of the Act. In this regard, the activities which Sentech may undertake in providing the carrier of carriers service to public switched telecommunications service licensees, mobile cellular telecommunications licensees and under-serviced area licensees will be curtailed as Sentech will not be able to provide a full service offering. This in turn will detrimentally impact on Sentech's ability to compete with other providers of international telecommunication services such as Telkom SA Limited ("Telkom") and the Second Network Operator ("SNO")

and will affect the viability and profitability of its licensed activities. These prejudicial and detrimental effects could never have been the intention of the legislature in granting the carrier of carriers licence to Sentech.

3 RELIEF SOUGHT BY SENTECH

- 3.1 In the circumstances, Sentech requests the Authority to make an order or a determination in terms of section 48(1)(e) read with sections 34(3), 34(4), 34(5) and 35 in respect of the carrier of carriers licence in the following terms:-
- 3.1.1 That clause 3.1 is amended by the addition of the word "lease" after the word "own" in the first line of clause 3.1.
- 3.1.2 That further and/or alternative relief is granted if deemed appropriate by the Authority.

DATED at SANDTON this _____ day of AUGUST 2002.

Dr. S Mokone -Matabane for Applicant

Octave Street
Honeydew

Tel: (011) 471-4476 Fax: (011) 471-4507

Ref: Dingane Dube/CoC/kbl/Authority

TO:

INDEPENDENT COMMUNICATIONS AUTHORITY OF SOUTH AFRICA Office of the Chairperson : Mandia Langa Block B, Pinmill Farm 164 Katherine Street Sandton

Received a copy hereof this day of AUGUST 2002

For: Independent Communications
Authority of South Africa

DRAFT

INDEPENDENT COMMUNICATIONS AUTHORITY OF SOUTH AFRICA

PROPOSED AMENDMENT TO:

LICENCE ISSUED TO SENTECH LIMITED TO PROVIDE AN INTERNATIONAL

TELECOMMUNICATION GATEWAY SERVICE ENABLING IT TO OPERATE AS A

CARRIER OF CARRIERS IN TERMS OF SECTION 32C(1)(A) AND SECTION 32C(6) OF

THE TELECOMMUNICATIONS ACT 103 OF 1996 AS AMENDED

INTERNATIONAL TELECOMMUNICATION GATEWAY SERVICE

TERMS AND CONDITIONS

1 DEFINITIONS

- 1.1 In this Licence and the annexes to this Licence:-
- 1.1.1 clause headings are for convenience and are not to be used in the interpretation of the Licence in which such headings appear;
- 1.1.2 unless the context indicates a contrary intention, an expression which denotes
- 1.1.2.1 any gender includes the other gender,
- 1.1.2.2 a natural person includes a juristic person and vice versa, and
- 1.1,2.3 the singular includes the plural and vice versa.
- 1.2 In this Licence, the following expressions bear the meanings assigned to them below and cognate expressions bear corresponding meanings -
- 1.2.1 "Authority" means the Independent Communications Authority of South Africa established in terms of section 3 of the Independent Communications Authority of South Africa Act, 2000 and its successors;
- 1.2.2 "Applicable Interest Rate" means the prime interest rate as published from time to time by the South African Reserve Bank, established by Section 9 of the Currency and Banking Act, No 31 of 1920, and operating in terms of the South African Reserve Bank Act, No 90 of 1989;
- 1.2.3 "Call" means a call (including, any Message conveyed by means of a telecommunication system) which originates in the telecommunication system of any Operator in the Territory, conveyed by the Licensee via the International Telecommunication Gateway Facility and which terminates in a telecommunication system in a country other than the Territory or vice versa or a call (including, any Message conveyed by means of a telecommunication system) which originates and terminates in a telecommunication system of an operator licensed to provide international services in a country other than the Territory which is conveyed via the International Telecommunication Gateway Facility on a wholesale basis;

- 1.2.4 "Carrier of Carriers" means a telecommunication service (including any signal conveyed by means of a telecommunication system of that service) which -
- 1.2.4.1 originates on the telecommunication system of a public switched telecommunication service licensee or mobile cellular telecommunication licensee or an under-serviced area licensee in the Republic and terminates in a telecommunication system in another country or vice versa; or;
- 1.2.4.2 originates and terminates in a telecommunication system of an operator licensed in another country to provide international services, but is conveyed via a telecommunication system in the Republic on a wholesale basis,

but which specifically excludes the termination of international telecommunication services to end-users directly in the Republic;

- 1.2.5 "Effective Date" means 7 May 2002 and more particularly, the commencement of business on that day;
- 1.2.6 "Financial Year" means the financial year end of the Licensee ending on 31 March of each and every year;
- 1.2.7 "Historically Disadvantaged Person" means, in relation to natural persons, persons who are generally regarded as South African Black persons (i.e. Africans, Coloureds or Indians), women and persons with disabilities and, in relation to juristic persons, means entities that are controlled by Historically Disadvantaged Persons, and Historically Disadvantaged Group shall have a corresponding meaning
- 1.2.8 "International Telecommunication Gateway Facility" means the telecommunication systems and telecommunication facilities which are owned, installed, maintained, operated and used by the Licensee for the purposes of providing the Carrier of Carriers service, by whatever means, including any telecommunication facility, by means of which Calls are conveyed between all or any of:-
- 1.2.8.1 1 (one) or more network connection points outside the Territory; and
- 1.2.8.2 1 (one) or more network connection points within the Territory;
- 1.2,9 "ITU" means the International Telecommunication Union;

- 1.2.10 "Licence" means this Licence issued by the Authority to the Licensee in terms of sub-sections 32C(1)(a) and 32C(6) of the Telecommunications Act;
- 1.2.11 "Licensee" means Sentech Limited, registration number 1990/001791/07, a company duly registered and incorporated under the company laws of the Territory;
- 1.2.12 "Licence Fee Income" means the total annual invoiced revenue of the Licensee (less discounts, value-added tax and other indirect taxes) derived from Operators for the provision to them of any aspect of the Carrier of Carriers service, less net interconnect fees and charges and bad debts incurred and as provided for in terms of the Income Tax Act, No 58 of 1962;
- 1.2.13 "Licence Period" means the term for which this Licence, including the applicable frequency spectrum licence/s, shall be in force (including such periods in respect of which this Licence and the applicable frequency spectrum licence/s may be renewed);
- 1.2.14 "Message" means any sound, signal, sign or image conveyed by means of the International Telecommunication Gateway Facility;
- 1.2.15 "Minister" means the Minister of Communications:
- 1.2.16 "Operators" means any public switched telecommunication service licensee or any entity licensed to provide similar services, mobile cellular telecommunication service licensee or any entity licensed to provide similar services or any Under-Serviced Area Licensee, licensed to provide such services in terms of the Telecommunications Act;
- 1.2.17 "Public Land Mobile Network" means the telecommunication systems used by mobile cellular telecommunication service licensees to provide mobile cellular telecommunication services;
- 1.2.18 "Public Switched Telecommunication Network" means the telecommunication systems used by public switched telecommunication service licensees to provide public switched telecommunication services;
- 1.2.19 "Regulations" means regulations promulgated in terms of section 96 of the Telecommunications Act No. 103 of 1996;
- 1.2.20 "Telecommunications Act" means the Telecommunications Act No 103 of 1996;
- 1.2.21 "Territory" means the Republic of South Africa;

- 1.2.22 "Under-Serviced Areas" means those geographic areas as determined by the Minister by notice in the Government Gazette as having a teledensity of less than 5%;
- 1.2.23 "Under-Serviced Area Licensee" means any person licensed to provide telecommunication services in terms of section 40A(2) of the Telecommunications Act;
- 1.2.24 "Universal Service Fund" means the fund established in terms of section 65 of the Telecommunications Act.
- 1.3 Words and expressions defined in any clause shall, for the purposes of that clause, bear the meaning assigned to such words and expressions in such clause.
- 1.4 If any provision is a substantive provision conferring rights or imposing obligations on the Licensee, notwithstanding that it is only in the interpretation clause, effect will be given to it as if it were a substantive provision in the body of the Licence.
- 1.5 All terms and expressions not expressly defined herein shall have the meanings assigned to them as set out in the Telecommunications Act, as amended, and all applicable Regulations issued in terms of that Act. The Licensee is required to comply with the Telecommunications Act and all applicable Regulations.

2 LICENCE PERIOD

The initial term of the Licence shall be for a period of 25 (twenty five) years from the effective date.

3 LICENCE AUTHORITY

- 3.1 The Licensee is authorised by this Licence to procure, own, <u>lease</u>. (<u>Sentech's request</u>) construct, maintain, operate and use an International Telecommunication Gateway Facility for the provision of a Carrier of Carriers service within and outside the Territory.
- 3.2 The Licensee shall be entitled and is authorised in terms of this Licence to:-
- 3.2.1 connect or cause to be allowed to be connected, to the International Telecommunication Gateway Facility:-

- 3.2.1.1 the Public Switched Telecommunication Networks:
- 3.2.1.2 the Public Land Mobile Networks,
- 3.2.1.3 any telecommunication system operated by an Under-Serviced Area Licensee;
- 3.2.1.4 any telecommunication system situated outside the Territory and licensed in another country to provide international telecommunication services except where such connection would contravene a treaty or other agreement to which the Territory is a party;
- 3.2.1.5 any telecommunication facility required for the provision of a Carrier of Carriers service;
- 3.2.2 convey to and from any connected telecommunication system as contemplated in clause 3.2.1 any Call and to perform any switching incidental to such conveyance;
- 3.2.3 provide the Carrier of Carriers service to any Operator and to use the Carrier of Carrier service for the Licensee's own purposes or business activities; and
- 3.2.4 provide any telecommunication service that is incidental to the service referred to in clause 3.2.3;
- 3.3 In addition to the rights referred to in clause 3.1, the Licensee shall be entitled to:-
- 3.3.1 procure, design, develop and maintain any software or hardware to be used in connection with any telecommunication system for the provision of its Carrier of Carriers service;
- 3.3.2 use any protocol application in its provision of the Carrier of Carriers service;
- 3.3.3 use its own telecommunications facilities in the provision of its Carrier of Carriers service and the interconnection of the same to any telecommunication system;
- 3.3.4 use any telecommunication facility for the connection of the International Telecommunication Gateway Facility to the telecommunication system of any Operator or any operator licensed to provide international telecommunication services outside of the Territory;
- 3.3.5 provide any service which is ancillary to the Licensee's provision of a Carrier of Carriers service, including billing and installation, maintenance and repair services;

- 3.3.6 enter into suitable arrangements with any licensed Operator for the sharing of infrastructure and facilities, for the co-location of facilities and infrastructure and to interconnect with any Operator or operator licensed to provide international telecommunication services outside of the Territory; and
- 3.3.7 use the frequency spectrum assigned by the Authority in the frequency spectrum licence for the provision of the Carrier of Carriers service.
- 3.4 The Licensee shall be entitled by virtue of this Licence to operate the International Telecommunication Gateway Facility and to provide a Carrier of Carriers service together with all or any other rights granted to the Licensee under this Licence, provided that any service which is ancillary to the Licensee's provision of Carrier of Carriers service, including billing and installation, maintenance and repair services, shall all be subject to the provisions of the Telecommunications Act, the relevant Regulations promulgated from time to time in terms of section 96 and the terms and conditions of this Licence.
- 3.5 The Licensee shall use all reasonable endeavours to ensure that its employees, agents and contractors comply with the provisions of this Licence in the exercise or performance of the Licensee's rights and obligations as set out in this Licence.

4. LICENCE FEES

4.1 Fixed Licence Fee

In consideration for the grant of the Licence, the Licensee shall pay to the Authority a fixed licence fee in respect of the initial term of the Licence, referred to in clause 2, in the amount of R25 (Twenty Five) Million Rand.

- 4.1.1 Should the Licensee fail to make payments of any amounts due in terms of clause 4.1.2 on the due date for payment thereof, then and in such event, interest at the Applicable Interest Rate shall accrue on the amount which is due in terms of clause 4.1.2., such interest to be calculated from the due date for payment thereof to the date of actual payment.
- Subject to clause 4.1.3, the principal amount of the fixed licence fee referred to in clause 4.1 shall be payable by the Licensee to the Authority in 5 (five) equal annual instalments, the first instalment to be paid on the third anniversary of the Effective Date.

- 4.1.3 The Licensee may accelerate payments of the fixed licence fee referred to in clause 4.1, by making payment prior to the dates specified in 4.1.2 and, in such cases, the provisions of clause 4.1.2 shall still apply.
- 4.1.4 In addition to the fixed licence fee, the Licensee shall pay the fees stipulated in its frequency spectrum licence and such other fees as may be payable in terms of such Regulations as may apply to it from time to time.

4.2 Annual Variable Licence Fee

- 4.2.1 Subject to what is stated below, an annual variable licence fee in an amount equal to 0.5% (one half of one percent) of the audited Licence Fee Income shall be payable by the Licensee to the Authority.
- 4.2.2 The first payment of the annual variable licence fee referred to in clause 4.2.1 shall be made within 3 (three) months of the Licensee's Financial Year in the third year of the Licence Period and shall be calculated on the basis of the Licence Fee Income generated during the third year of the Licence Period.
- 4.2.3 After the fourth year of the Licence Period, the annual variable licence fee shall be payable within ninety (90) days after the expiry of Sentech's financial year in question.
- 4.2.4 The annual variable licence fee payable by the Licensee shall be reviewable by the Authority after the fifth anniversary of the Effective Date, provided that, in the event of the percentage of Licence Fee Income payable by the Licensee being increased by the Authority, the percentage so increased shall not exceed the percentage of net operating income payable by similarly licensed operators in respect of the corresponding licence fees payable by them.
- 4.3 In reviewing the percentage of Licence Fee Income payable by the Licensee in terms of this clause, the Authority shall have due regard to factors such as the number of similarly licensed operators in the Territory, the spectrum allocated to such similarly licensed operators and the percentage of net operating income payable by such similarly licensed operators.
- 4.4 The provisions of clause 4.2 shall apply *mutatis mutandis* to all subsequent Licence Periods of this Licence.

A CONTRACTOR OF THE STATE OF TH

Control of the second states

5 UNIVERSAL SERVICE OBLIGATION

The Licensee shall pay annual contribution to the Universal Service Fund as prescribed by regulation in terms of section 67(2) of the Telecommunication Act.

6 PERFORMANCE SPECIFICATIONS

- 6.1 The Carrier of Carriers service provided by the Licensee shall conform to such Regulations as prescribed by the Authority.
- 6.2 The Licensee shall introduce measures and at all times use its resources to ensure that any telecommunication apparatus used in connection with the International Telecommunication Gateway Facility or in the provision of the Carrier of Carriers service, including any telecommunication facilities owned by the Licensee shall comply with:
- 6.2.1 the requirements for electromagnetic compatibility; and
- 6.2.2 the requirements for network interoperability,

as prescribed by the Authority in accordance with international specifications and standards.

7 FAIR TRADING

The Licensee shall not demonstrate any undue preference to or exercise any undue discrimination against any Operator in respect of the provision of the Carrier of Carriers service or in respect of the construction or maintenance of any telecommunication facility or connection to the Licensee's International Gateway Telecommunication Facility.

8 ACCOUNTS

- 8.1 The Licensee shall maintain proper accounting records in a form, which is sufficient to show and explain its transactions and in this regard, the Licensee shall maintain such records as fairly represent the cost, revenue and financial position of the Licensee's business activities in terms of this Licence.
- 8.2 Within 3 (three) months of the end of each Financial Year of the Licensee, the Licensee shall present to the Authority audited annual financial statements

prepared in accordance with Generally Accepted Accounting Practice and as approved by the Licensee.

9 TRANSFER OF LICENCE AND OWNERSHIP AND CONTROL

9.1 Transfer of Licence

3.2 The Licensee shall not transfer the Licence without the prior written consent of the Authority having been obtained in terms of the Telecommunications Act and the applicable Regulations.

9.2 Transfer of ownership and control

- 9.2.1 Any transfer of shares of the issued share capital of the Licensee being acquired by any party which is not a shareholder of the Licensee as at the Effective Date shall require prior written approval by the Authority.
- 9.2.2 Any transaction, agreement or undertaking entered into by the Licensee with any other party that could have the effect of changing the control of the Licensee, shall require prior written approval by the Authority.

10 EMPLOYMENT EQUITY, HUMAN RESOURCE DEVELOPMENT AND TRAINING

- 10.1 The Licensee shall comply with the provisions of the Employment Equity Act, No 55 of 1998, and the Skills Development Act, No 97 of 1998, in relation to employment equity, human resource development and training. All reports submitted to the Department of Labour in this regard by the Licensee shall also be submitted to the Authority.
- 10.2 The Licensee shall participate in the development of the telecommunication industry by, inter alia:
- 10.2.1 supporting independent contractors from Historically Disadvantaged Groups; and
- 10.2.2 supporting industry development initiatives.

11 GENERAL

- 11.1 Should any provision of this Licence be invalid and unenforceable for any reason, the remaining provisions shall nevertheless remain of full force and effect.
- 11.2 To the extent that this Licence may be inconsistent with any provisions of the Telecommunications Act or any applicable Regulations, such provisions of the Telecommunications Act or the Regulations shall take precedence over this licence.

12 CONFIDENTIALITY OF INFORMATION

- 12.1 The Licensee shall not disclose the content of any signal transmitted or intended to be transmitted over its International Telecommunication Gateway Facility, unless required to do so by a court order or in terms of any law.
- 12.2 The Licensee shall not use any information regarding its past, current or potential customers for purposes other than those for which the information was obtained, unless the customer gives prior written consent to such other use.

13 RENEWAL OF THE LICENCE

The Licence may be renewed in terms of the Telecommunication Act.

14 CANCELLATION OF LICENCE

In the event of the liquidation of the Licensee, the Licensee shall notify the Authority in writing of the cancellation of the Licence.

15 REVOCATION OF THE LICENCE

The Licence may be revoked in accordance with the Telecommunications Act.

16 FORCE MAJEURE

16.1 The Licensee shall not be liable for any default or delay in the performance of its obligations under the Licence if and to the extent:-

- 16.1.1 such default or delay is caused, directly or indirectly, by fire, flood, earthquake, elements of nature or acts of God, riots, civil disorders, rebellions or revolutions in any country or any other cause beyond the reasonable control of the Licensee; and
- 16.1.2 the Licensee is without fault in causing such default or delay.

17 NOTICES

· 14. *

17.1 The Licensee selects as its domicilia citandi et executandi the following physical addresses:-

<u>Name</u>	Physical Address	<u>l eletax</u>			
Sentech Limited	Sender Technology Park	(011) 471-4753			
	Octave Road Radiokop				

or such other address or telefax number as may be substituted by notice given as herein required. The Licensee shall be entitled from time to time by written notice to the Authority, to vary its *domicilium* to any other address within the Territory.

- 17.2 Any notice addressed to the Licensee at its physical address shall be delivered by hand, or sent by telefax.
- 17.3 Any notice shall be deemed to have been given:-
- 17.3.1 if hand delivered during business hours on a business day, on the day of delivery;
- 17.3.2 if sent by telefax, on the date of sending of such telefax.

18 AMENDMENT

- 18.1 Subject to the provisions of section 48 of the Telecommunications Act, the Authority shall be entitled to amend any term or condition of this Licence.
- No amendment, alteration, addition, variation or consensual cancellation of this Licence shall be of any force or effect unless reduced to writing and signed by the Authority.

19 MISCELLANEOUS PROVISIONS

19.1 Adherence to International Standards

19.1.1 The Licensee shall comply with the applicable standards and requirements of the ITU, as agreed to or adopted by the Republic.

19.2 Legal Compliance

19.2.1 The Licensee shall comply with all applicable laws of the Republic and with all international obligations to which the Republic is bound and, unless expressly indicated to the contrary, nothing in this Licence shall be construed or understood so as to relieve or exempt the Licensee from complying with the provisions of this Licence, the Telecommunications Act or any other law to the extent that it may be applicable.

19.3 Access to network facilities

19,3.1	Upon reasonable notice, the Licensee	shall	provide	access	to	its	network
0	facilities for inspection by the Authority.	2.7					

Issued at				_ on this		(day) /	(moi	(month) / 20		
Y P	2	33,000	81	# Fr 10						
			100	× * 1		100	Y PE Y		82	
	727		0.3	. FO 640		54	# K K K T N	25		
MANDLA LAN	GA	7.00			* * * * *	1 - 1	" - 174 - 184 - " 374	a i	50 St	
CHAIRPERSON	1			500 60 500	w = 50	100 g = 10	1.5			
INDEPENDENT	20	NICATIO	NS A	UTHORIT	Y OF S	OUTH AF	RICA (ICA	SA)	n 72	
w	. #	4			25 1.		va	eroa E		

NKATEKO NYOKA
CHIEF EXECUTIVE OFFICER
INDEPENDENT COMMUNICATIONS AUTHORITY OF SOUTH AFRICA (ICASA)

BEFORE THE INDEPENDENT COMMUNICATIONS AUTHORITY OF SOUTH AFRICA ("THE AUTHORITY")

The second of the second of the second

For Fishing Parks of

In the ex parte matter of:

SENTECH LIMITED

Applicant

APPLICATION IN TERMS OF SECTION 48(1)(e) IN RESPECT OF SENTECH'S MULTIMEDIA SERVICES LICENCE

4 THE APPLICANT

- 4.1 The Applicant is Sentech Limited ("Sentech"), a company established pursuant to the Sentech Act 63 of 1996 and incorporated in accordance with the company laws of the Republic of South Africa and having its principal place of business and registered office at Octave Street, Radiokop, Honeydew, Gauteng Province, South Africa.
 - 4.2 Sentech is the holder of a licence to provide multimedia services in terms of section 32C (1)(b) ("the multimedia services licence") of the Telecommunications Act 103 of 1996 ("the Act"). The multimedia services licence was issued to Sentech by the Independent Communications Authority of South Africa ("the Authority") on 7 May 2002 in terms of section 32C(6) of the Act and was also subsequently published under Notice 686 of 2002 in Government Gazette No. 23405 dated 8 May 2002.
 - 4.3 Sentech is also the holder of a licence to provide an international telecommunication gateway service enabling it to operate as a carrier of carriers in terms of section 32C (1) (a) ("the carrier of carriers licence") of the Act. The carrier of carriers licence was issued to Sentech by the Authority in terms of section 32C (6) of the Act on 7 May 2002 and was subsequently published under Notice 684 of 2002 in Government Gazette No. 23405 dated 8 May 2002.

5 THE APPLICATION

Sentech has brought this application in terms of section 48(1)(e) of the Act, as it has three major concerns with the provisions of its multimedia services licence. As a result of these concerns Sentech, requires the multimedia services licence to be amended by the Authority as set out in this application. The amendments, which Sentech requires to its multimedia services licence and its motivations for requiring the same are set out below.

6 CLAUSE 3.6 TO BE DELETED IN ITS ENTIRETY

6.1 Clause 3.6 to the multimedia services licence provides:-

"Nothing contained in this Licence shall authorise the Licensee to provide telecommunication facilities to anyone licensed under the Telecommunications Act to provide value added network services, in circumstances where that person, by obtaining the telecommunications facilities in question, from the Licensee, violates a term of its own telecommunication licence or otherwise acts in a manner inconsistent with that contemplated by section 40(2) of the Telecommunications Act."

6.2 By including clause 3.6 in the multimedia services licence, the Authority has in effect made Sentech subject to the provisions of section 40(2) to the Act. The Authority's conduct in making Sentech subject to the provisions of section 40(2) of the Act is entirely inappropriate and contrary to the statutory rights and entitlements conferred on Sentech in its provision of multimedia services. As such, it is submitted that the Authority has acted ultra vires the powers conferred on it in terms of the Act. Sentech's reasons for this contention are based on the following:-

6.2.1 Section 40(2) of the Act provides:-

"A licence to provide any value- added network services, including, but not limited to, electronic data interchange, electronic mall, protocol conversion, access to database or a managed data network service, shall contain a condition that the service in question be provided by means of telecommunication facilities until-

- (a) 7 May 2002, provided by Telkom or made available to Telkom as contemplated in section 44; and
- (b) after 7 May 2002, provided by Telkom and the second national operator or any of them until a date fixed by the Minister by notice in the Gazette."

- 6.2.2 Section 40(2) requires all licences issued to value added service licensees ("VANS") to contain a condition that services provided by VANS are to be provided by means of telecommunication facilities provided by Telkom SA Limited ("Telkom") and the Second Network Operator ("SNO"), once licensed. As Sentech has not been granted a licence to provide VANS, but is licensed to provide multimedia services, the provisions of section 40(2) should not apply to it and there is no statutory basis for including a provision in the multimedia services licence which purports to draw Sentech into the provisions of section 40(2). Thus, there is a concern that the Authority has acted ultra vires the powers conferred on it in terms of the Act by imposing this condition on Sentech under circumstances where Sentech is neither a VANS nor subject to the provisions of section 40(2).
- 6.2.3 In addition, section 40(2)(b) provides that, until a date fixed by the Minister, only Telkom and the SNO shall be entitled to provide telecommunication facilities to VANS. This provision must be viewed in the context of the various debates which took place in regard to whether a policy of service based or facilities based competition should be followed, post Telkom's exclusivity. Ultimately, it was decided by the Minister of Communications ("the Minister") to follow a policy which would give effect to the stated intention of managing the liberalisation of the telecommunications sector by first introducing service based competition to be followed by the introduction of facilities based competition at a later date. It is precisely this policy decision which is given effect to in section 40(2)(b) as the Minister will determine the date upon which other entities will be entitled to supply telecommunication facilities to VANS.
- 6.2.4 Notwithstanding the provisions of section 40(2)(b), the right granted to Telkom and the SNO does not preclude Sentech from providing a multimedia service to VANS by means of its own facilities as no similar obligation is imposed on Sentech in terms of section 32C or in any other section to the Act. This is borne out by the fact that in terms of section 40(2) to the Act, it is a VANS service and only a VANS service, which is to be provided by means of telecommunication facilities provided by Telkom or the

SNO. No similar restriction is contained in the Act in respect of a multimedia service. Thus, in terms of the Act, Sentech in providing multimedia services to a VANS, is entitled to provide its own telecommunications facilities for the provision of such services up to and including the point of interconnection. Once multimedia traffic has been passed by means of interconnection from Sentech to the network of a VANS, such traffic will then be carried by the VANS network. It is only the VANS and not Sentech's network, which is required to be comprised of telecommunication facilities provided by Telkom or the SNO.

- 6.2.5 The provisions of clause 3.6 to the licence are far wider in application than the condition, which in terms of section 40(2) is to be included in all VANS licences. This is due to the fact that Sentech is required in terms of clause 3.6 to ensure that in providing multimedia services to VANS, that there is no violation by a VANS of its own licence and that the provisions of section 40(2) of the Act are not contravened. In fact, the obligation imposed on Sentech in terms of clause 3.6 is so wide, that Sentech may well be liable for ensuring that all activities conducted by VANS when receiving a multimedia service from Sentech, do not violate the terms and conditions of their licence or the Act. The over broad effects of this clause are such that it will be impossible for Sentech to actually adhere to the provisions of the same. The effect of this will be severely prejudicial to Sentech in that VANS providers will probably be discouraged from obtaining multimedia services from Sentech for fear that the acquisition of such services may violate the conditions of their licences. Similarly, Sentech will bear a substantial risk in providing such services to VANS as should any VANS provider use its telecommunication facilities, contrary to the provisions of the Act, Sentech may well be in breach of its licence.
- 6.2.6 Insofar as clause 3.6 in effect prohibits Sentech from providing services to VANS, this is contrary to the common law where it has long been recognised that where an empowering Act authorises regulation and control, this does not include the power to prohibit [see R v Williams 1914 AD 460; S v Perumal 1977 (1) SA 526 (N); Moekena v Commissioner of Prisoners 1985 (1) SA 377

(A); City of Cape Town v Ad Outpost (Pty) Ltd 2000 (2) BCLR 130 (C)]. As there is no limitation imposed on Sentech in respect of the provision of a multimedia services to VANS, the consequences of clause 3.6, which will be to prohibit the provision by Sentech of multimedia services to VANS will be contrary to the common law.

- 6.3 Clause 3.6 was not included in the draft multimedia services licence published for public comment under Notice 2405 of 2001 in Government Gazette No. 22976 dated 20 December 2001, nor was clause 3.6 included in the final draft multimedia services licence handed to Sentech for comment by the Authority on 30 April 2002. In fact, notwithstanding the public process, clause 3.6 to the multimedia services licence only came to Sentech's attention on the publication of the final licence in the Government Gazette under Notice 686 of 2002 on 8 May 2002
- The Promotion of Administrative Justice Act gives effect to the Constitutional right to administrative justice. Section 3(1) of this Act states "[a]dministrative action which materially and adversely affects the rights or legitimate expectations of any person must be procedurally fair". As to the content of the requirement to act fairly, the court in Du Preez v Truth and Reconcilliation Commission [1997 (3) SA 204 (A)] quoted with approval the guidelines laid down in the English case of Doody v Secretary of State for Home Development and other Appeals [(1993) 3 All ER 92 (HL) 106], which held that "fairness will very often require that a person who may be adversely affected by the decision will have an opportunity to make representations on his own behalf either before the decision is taken with a view to producing a favourable result, or after it is taken, with a view to procuring its modification, or both".
- 6.5 The Act itself provides for such a process of notification and comment, and it would have been procedurally fair for Sentech to have been given an opportunity to comment on clause 3.6 prior to the insertion of the same in the final multimedia services licence as clause 3.6 clearly affects Sentech's rights under section 32C to provide multimedia services. The fact that the provisions

of 3.6 was not included in the draft licence certainly gave rise to a legitimate expectation that Sentech would be entitled to provide facilities to VANS, which constitutes a significant market for Sentech's multimedia services. It is therefore submitted that the insertion of clause 3.6 constitutes an invalid administrative act on the part of the Authority.

In summation, the Authority's inclusion of clause 3.6 in the multimedia services licence is *ultra vires* the powers conferred upon the Authority in terms of the Act. Further, the inclusion of clause 3.6 constitutes conduct which is unreasonable and unjustifiable when one has regard to the requirements of section 35(5) to the Act, which obliges the Authority to ensure that licences are granted on conditions appropriate to the licence and consistent with the objects of the Act. Furthermore, the Authority's failure to allow Sentech to comment on clause 3.6 and its late inclusion in the multimedia services licence constitutes a procedural irregularity. It is for all these reasons that Sentech requires clause 3.6 to be deleted from the multimedia services licence in its entirety.

7 CLAUSE 1.2.18 TO BE DELETED IN ITS ENTIRETY

- 7.1 Clause 1.2.18 sets out the definition of an operator as this term is to be understood and used in terms of the multimedia services licence and provides as follows:-
 - " 'Operator' means any Public Switched Telecommunication Service Licensee or any entity licensed to provide similar services, Mobile Cellular Telecommunication Service Licensee or any entity licensed to provide similar services or any Under-Serviced Area Licensee, licensed to provide such services in terms of the Telecommunications Act."
- 7.2 The term "Operator" as defined, is referred to in clause 3.3.5 to the multimedia services licence. Clause 3.3.5 entitles Sentech to enter into suitable arrangements with only those operators as defined in clause 1.2.18 for the sharing of infrastructure, the co-location of facilities and for the purposes of interconnecting with such operators. The definition of "Operator" when read in conjunction with the provisions of clause 3.3.5 to the licence places an unwarranted restriction on Sentech's ability to contract freely with third parties for the sharing of infrastructure, co-location and interconnection.

- 7.3 There is no statutory basis for the imposition of such a limitation on Sentech and Sentech should not be constrained in its ability to enter into suitable arrangements for infrastructure sharing and co-location. Other licensees are entitled to enter into infrastructure sharing and co-location arrangements with other telecommunication service licensees and third parties. Any attempt to limit Sentech's rights in this regard where there is no statutory justification for such a limitation amounts to no more than arbitrary and discriminatory conduct on the part of the Authority.
- There is also a concern that the provisions of clause 1.2.18 when read with 7.4 clause 3.3.5 are contrary to the provisions of section 43 to the Act. The effect of clause 3.3.5 when read with clause 1.2.18 is that Sentech is only permitted to interconnect with public switched telecommunications service licensees, mobile cellular telecommunication licensees and under-serviced area licensees. On the other hand, section 43(1)(a) to the Act places a positive obligation on public telecommunications service licensees to interconnect their switched telecommunication systems with the telecommunication system of "any other person providing telecommunication services" when requested to do so. One of the fundamental principles in respect of interconnection is that it will be available to any person providing telecommunications services, there only being a positive obligation on public switched telecommunications service licensees to provide interconnection. Thus, it is entirely contrary to the provisions of section 43 to limit Sentech's ability to interconnect with only those entities listed within the definition of an "Operator" in clause 1.2.18.
- 7.5 The exact same definition of an "Operator" is found in clause 1.2.16 of the carrier of carriers licence. The use of such a definition in the carrier of carriers licence is justified due to the limitation contained in the definition of "carrier of carriers" in the Act which obliges Sentech to only carry international traffic which originates on the telecommunication system of a public switched telecommunications service licensee, mobile cellular telecommunication service licensee or an under-serviced area licensee within the Republic. There is however no such similar limitation or obligation imposed on Sentech in respect of its provision of multimedia services and it is

inappropriate for such a limitation to be imposed on Sentech in terms of its multimedia service licence. As the provisions of clause 1.2.18 read with clause 3.3.5 are contrary to the provisions of the Act, the Authority, by imposing such a condition on Sentech has acted outside of the powers conferred on it in terms of the Act.

Once again, the inclusion of the definition of "Operator" in clause 1.2.18 in the multimedia services licence when read with clause 3.3.5, constitutes conduct on the part of the Authority which is ultra vires the powers conferred on it in terms of the Act. The Authority's conduct is also arbitrary and discriminatory in that no similar limitation is imposed on other licensees. In the circumstances, Sentech requires clause 1.2.18 to be deleted in its entirety and for the word "Operator" in clause 3.3.5 to be amended and substituted with the word "operator".

8 CLAUSE 5.2 AND ANNEXURE A TO BE DELETED IN THEIR ENTIRETY

- 8.1 The multimedia services licence is a unique licence and it is the first licence of its kind to be issued worldwide. The demand for multimedia services and the ultimate profitability of this licence is difficult to estimate and as such, the multimedia services licence should be approached from a 'green fields' perspective. Whilst Sentech supports and endorses the objectives underlying universal service, access and affordability and is willing to contribute to the Universal Service Fund on a fair and equitable basis, Sentech is of the view that the imposition of community service obligations must be viewed in the context of the existent universal service framework and the rights and limitations of the multimedia services licence.
- 8.2 In terms of clause 5.2 to the multimedia services licence, Sentech is required to establish and maintain school Internet laboratories in 500 (five hundred) schools in rural areas over a period of 5 (five) years. The specifics of Sentech's community service obligations are detailed in Annexure "A" to the licence and include, inter alia, the provision of local area networks, Internet connectivity, work stations with enhanced functionality and furniture. In addition, Sentech is required to refurbish and upgrade buildings where appropriate and to provide secure and useable computer laboratories. Technical training and professional

development opportunities also form part of Sentech's community service obligations.

- It is clear that the community service obligations in the multimedia services licence will require Sentech to provide certain services that fall outside of its licensed role as a provider of multimedia services. It is submitted that in granting a licence subject to conditions, such conditions should properly relate only to the activities to be undertaken pursuant to the licence. It is not competent for the Authority to impose obligations which are outside of the licensed activities, such as the refurbishment and upgrading of buildings, the establishment and procurement of computer laboratories, work stations, furniture etc.
- In addition, section 5 of the Sentech Act 1996 (Act 63 of 1996) specifically states that "the main object and main business of the Company shall be to provide (a) as a common carrier, broadcasting signal distribution for broadcasting licensees in accordance with the provisions of the Independent Broadcasting Authority Act; and (b) telecommunications services in accordance with the provisions of the Telecommunications Act, 1996". Insofar as certain of the community service obligations imposed under the multimedia services license require Sentech to engage in activities materially distinct from those it is empowered to do in terms of its enabling Act, the Authority is in effect obliging Sentech to function beyond its objects, which would render such acts ultra vires.
- 8.5 It is submitted that the community service obligations that Sentech is required to provide in terms of its multimedia services licence and which fall outside of its licensed activities, should properly be undertaken under the auspices of the Universal Service Agency, to be subsidised through the Universal Service Fund as envisaged in section 66(1) of the Act. For the multimedia services licence to purport to make Sentech financially responsible for the facilitation and subsidising of a scheme that in terms of the Act is reserved for the Universal Service Agency (and ultimately, in terms of section 64, the Authority itself) amounts to an attempt to superimpose the duties of the Agency onto Sentech,

in dereliction of the Agency's role under the Act. It is respectfully submitted that this is ultra vires the powers of the Authority and fails to appreciate the purpose of Chapters XII and XIII of the Act as being the proper mechanism to facilitate and subsidise the twin ideals of universal access and service.

- In support of what is stated in paragraph 8.5, section 67(2)(a) to the Act 8.6 provides that the Authority is entitled to make regulations on "the basis and manner of determination of such contributions, which shall not exceed 0.5% of a licensee's annual turnover". Thus, the formula to be used by the Authority in determining universal service fund contributions is to be based on a percentage of a licensee's annual turnover. No mention is made of the imposition of community service obligations, which in application will amount to no more than a species of contribution to the universal service fund. As the Authority's powers in determining universal service fund contributions is limited to a percentage of a licensee's annual turnover, it is submitted, that the Authority is not empowered to impose community service obligations on licensees and is only empowered to impose an obligation to contribute to the universal service fund in accordance with the mechanism set out in section 67(2)(a) to the Act.
- Sentech has undertaken a cost analysis in respect of each and every item comprising its community service obligations and has estimated that the total costs of complying with its community service obligations will amount to the payment of R537 million over the 5 (five) year period. At this stage, the expected monetary returns from the multimedia services licence will not be sufficient to cover the fulfillment of this obligation and Sentech will have to borrow funding in order to comply with its community service obligations. In effect, the community service obligations imposed on Sentech will mean that Sentech will commence its provision of multimedia services from a deficit position in that it will have to obtain outside funding to meet these obligations. This in turn will create the very real risk that Sentech will not be able to attract other investors due to the high debts, which would be incurred by it in adhering to its community service obligations. It could never have been the intention of the Authority, in imposing these obligations on Sentech, to have created a

situation where at the outset, the whole viability and profitability of the multimedia services licence is compromised by the onerous burden placed on Sentech in respect of its community service obligations.

- 8.8 It is submitted that the unintended consequence of the inclusion of the community service obligations in the multimedia services licence will be to materially affect Sentech's financial status. This will have a detrimental effect on Sentech's profitability as well as its ability and incentive to reinvest capital in an industry that demands considerable expenditure in research and development. This is in apposition to Sentech's right under section 33(1) of the Constitution of the Republic of South Africa, Act 108 of 1996 ("the Constitution") to administrative action, which is lawful, reasonable and procedurally fair. The Authority is respectfully referred to the following recent judgements in support of Sentech's contention:-
- 8.8.1 In Marais v Interim Nasionale Mediese en Tandheelkundige Raad van Suid Afrika [(1997) 4 All SA 260 (O)], the court stated that if the decision was "grossly unreasonable to so striking a degree as to warrant the inference of a failure to apply his mind" that the decision of the respondent should be set aside.
- 8.8.2 In SARFU v President of the RSA [1998 (10) BCLR 1256 (T) 130 F-G], the relevant conduct was described as "so grossly unreasonable [as] to warrant the inference that [he] failed to apply his mind to the relevant issues in accordance with the behests of the statute and the tenets of natural justice".
- 8.8.3 In Roman v Williams [1998 (1) SA 270 (C) 281 D-E], Van Deventer J held that the constitutional right to justifiable administrative action "imports the requirement of proportionality between means and end and ... the role of the courts in judicial reviews is no longer limited to the way in which an administrative decision was reached but now extends to its substance and merits". Administrative conduct must therefore meet the requirements of suitability, necessity and proportionality in order to qualify as justifiable.

- 8.8.4 In Carphone (Pty) Ltd v Marcus [1998 (10) BCLR 1326 (LAC)] Froneman DJP stated that section 33(1) of the Constitution "introduces a requirement of rationality in the merit or outcome of the administrative decision".
- 8.9 In light of the above, it is submitted that the decision of the Authority to impose community service obligations of so onerous a nature is out of proportion to the purpose of such obligations, which purpose will be sufficiently served in a more justifiable manner through the mechanism of the Universal Service Agency and Fund, provided that all licensees are required to contribute thereto in a fair, equal and non-discriminatory manner. Furthermore, it is submitted that the consequences of the decision to impose such onerous obligations are so unreasonable vis a vis Sentech, that the inference might be made that the Authority failed to apply its mind to the import of imposing such community service obligations on Sentech and that such decision therefore falls short of the level of rationality required for a valid administrative act. Accordingly, Sentech urges the Authority to reconsider the merits of the community service obligations imposed on Sentech in terms of the multimedia services licence by deleting clause 5.2 and Annexure "A" in their entirety from the multimedia services licence.
- 8.10 The deletion of clause 5.2 and Annexure "A" from the multimedia services licence, will not result in Sentech not contributing to the objectives of universal access and service. Sentech reiterates that it is committed to the objectives of universal access, service and affordability and would be more than prepared to pay a percentage of turnover to the Universal Service Fund provided that the percentage is uniformly imposed on all licensees and provided that all of Sentech's community service obligations are removed from the licence. Sentech believes that the requirement that all telecommunication service licensees pay a uniform fixed percentage of the turnover to the Universal Service Fund will ensure that the objectives of universal service and access are achieved on an equal and non-discriminatory basis. This in turn will ensure the rapid fulfilment of these objectives as the universal service agency will at all times be able to monitor and determine the way in which funds are to be

deployed in order to best give effect to the objectives underpinning universal service.

to english at least ago tak £160 to 10

8.11 It is submitted that the retention of the community service obligations in the multimedia services licence will defeat the reasons for the legislature having granted the licence to Sentech and the objects of section 2 to the Act in that the continued viability and profitability of the licence will be seriously jeopardised and the many benefits to be derived from the licence will be lost. Accordingly, Sentech requests that the Authority delete clause 5.2 and Annexure "A" of the multimedia services licence in their entirety.

9 RELIEF SOUGHT BY SENTECH

9.1 In the circumstances, Sentech requests the Authority to make an order or a determination in terms of section 48(1)(e) read with sections 34(3), 34(4), 34(5) and 35 in respect of the multimedia services licence in the following terms:-

er stati yak yw a 💉

- 9.1.1 That clause 3.6 is deleted in its entirety.
- 9.1.2 That clause 1.2.18 is deleted in its entirety.
- 9.1.3 That the word "*Operator*" in clause 3.3.5 be amended and substituted with the word "*operator*".
- 9.1.4 That clause 5.2 is deleted in its entirety.
- 9.1.5 That Annexure "A" is deleted in its entirety.
- 9.1.6 That further and/or alternative relief is granted if deemed appropriate by the Authority.

of the first of the state of the state of

and the second of the second of the second

THE PARTY OF THE PARTY OF THE PARTY OF THE

n	ATED at	SANDTON this	day of	AUGUST 2002	١
U	AI ED al	SAND I UN IIIIS	uav ui	400001 2002	2,

Dr S Mokone-Matabane for Applicant Octave Street

Honeydew

Tel: (011) 471-4476

Fax: (011) 471-4507

Ref: D Dube/MM/kbl/02/Authority

TO:

INDEPENDENT COMMUNICATIONS AUTHORITY OF SOUTH AFRICA Office of the Chairperson : Mandla Langa Block B, Pinmill Farm 164 Katherine Street Sandton

Received a copy hereof this day of AUGUST 2002

For: Independent Communications Authority of South Africa

DRAFT

PROPOSED AMENDMENTS TO:

LICENCE ISSUED TO SENTECH LIMITED TO PROVIDE MULTIMEDIA SERVICES IN

TERMS OF SECTION 32C (1) (B) AND SECTION 32C (6) OF THE

TELECOMMUNICATIONS ACT 103 OF 1996 AS AMENDED

MULTIMEDIA SERVICE LICENCE

TERMS AND CONDITIONS

1. DEFINITIONS

- 1.1 In this Licence and the annexes to this Licence:-
- 1.1.1 clause headings are for convenience and are not to be used in its interpretation;
- 1.1.2 unless the context indicates a contrary intention, an expression which denotes:-
- 1.1.2.1 any gender includes the other gender;
- 1.1.2.2 a natural person includes a juristic person and vice versa,
- 1.1.2.3 the singular includes the plural and vice versa.
- 1.2 In this Licence, the following expressions bear the meanings assigned to them below and cognate expressions bear corresponding meanings —
- 1.2.1 "Applicable Interest Rate" means the prime interest rate as published from time to time by the South African Reserve Bank, established by Section 9 of the Currency and Banking Act, No 31 of 1920, and operating in terms of the South African Reserve Bank Act, No 90 of 1989;
- 1.2.2 "Approved Equipment" means telecommunication equipment which has been approved by the Authority under Chapter VI of the Telecommunications Act;
- 1.2.3 "Authority" means the Independent Communications Authority of South Africa established in terms of section 3 of the Independent Communications Authority of South Africa Act no. 13 of 2000, and its successors;
- 1.2.4 "Communication Configuration" means (i) point to point; or (ii) point to multipoint; or (iii) multipoint to point; or (iv) multipoint to multipoint; interactive transmission and distribution of Multimedia Services;
- 1.2.5 "Effective Date" means 7 May 2002 and more particularly, the commencement of business on that day;

- 1.2.6 "Financial Year" means the financial year of the Licensee ending on 31 March of each and every year;
- 1.2.7 "Historically Disadvantaged Person" means, in relation to natural persons, persons who are generally regarded as South African Black persons (i.e. Africans, Coloureds or Indians), women and persons with disabilities and, in relation to juristic persons, means entities that are controlled by Historically Disadvantaged Persons, and Historically Disadvantaged Group shall have a corresponding meaning;
- 1.2.8 "ITU" means the International Telecommunication Union;
- 1.2.9 "Licence" means this Licence issued by the Authority to the Licensee in terms of section 32C(1)(b) and 32C(6) of the Telecommunications Act;
- 1.2.10 "Licence Fee Income" means the total annual invoiced revenue of the Licensee (less discounts, value-added tax and other indirect taxes) derived from customers for the provision to them of any aspect of the Multimedia Service, less net interconnect fees and charges and bad debts incurred and as provided for in terms of the Income Tax Act, No 58 of 1962;
- 1.2.11 "Licence Period" means the term, referred to in clause 2, for which this Licence, including the associated frequency spectrum licence/s, shall be in force (including such periods in respect of which this Licence and the associated frequency spectrum licence/s may be renewed and any further periods of renewal);
- 1.2.12 "Licensee" means Sentech Limited, registration number 1990/001791/07, a company duly registered and incorporated under the company laws of the Territory;
- 1.2.13 "Minister" means the Minister of Communications;
- 1.2.14 "Multimedia Services" means a telecommunication service that integrates and synchronises various forms of media to communicate information or content in an interactive format, including services such as-

- a. internet through television;
- b. pay-per-view;
- c. video on demand;
- d. electronic transactions (including e-commerce);
- e. text;
- f. data;
- g. graphics;
- h. animation;
- audio;
- i. visual content,

But shall not include mobile cellular telecommunication services and public switched telecommunication services;

- 1.2.15 "Multimedia Service Network" means the Multimedia Service Network including any return path owned, constructed, maintained and operated by the Licensee for the purposes of providing Multimedia Services by whatsoever means through which Multimedia Traffic is conveyed in accordance with any Communication Configuration through the application of any applicable protocol between:
 - · 1 (one) or more terminal connection points;
 - 1 (one) or more network connection points;
 - A terminal connection point and a network connection point;
 - A terminal connection point or a network connection point, as the case may be and a corresponding point in another country;
- 1.2.16 "Multimedia Traffic" includes any text, data, graphics, audio, animation or visual content conveyed by the Licensee by means of its Multimedia Service Network.
- 1.2.17 "Network Connection Equipment" means in relation to the Multimedia Service Network or Multimedia Services, an item of telecommunication apparatus comprised in that telecommunication system which is not Terminal Connection Equipment and which is used or is meant to be used to provide interconnection with any other telecommunication system;

- 1.2.18 "Operator" means any Public Switched Telecommunication Service Licensee or any entity licensed to provide similar services, Mobile Cellular Telecommunication Service Licensee or any entity licensed to provide similar services or any Under-Serviced Area Licensee, licensed to provide such services in terms of the Telecommunications Act; {SENTECH requests that this clause be deleted in its entirety}
- 1.2.19 "PSTS" means public switched telecommunication service;

4 1 Y 1 Y

- 1.2.20 "Regulations" means regulations promulgated in terms of section 96 of the Telecommunications Act No. 103 of 1996;
- 1.2.21 "Telecommunications Act" means the Telecommunications Act No 103 of 1996:
- 1.2.22 "Terminal Connection Equipment" means in relation to the Multimedia Service

 Network an item of telecommunication apparatus comprised in that
 telecommunication system which is fixed or mobile and which enables:-
- 1.2.22.1 User Terminal Devices to be connected to the Multimedia Service Network;
- 1.2.22.2 Multimedia Traffic to be conveyed bidirectionally between the User Terminal Devices and the Multimedia Service Network; and
- 1.2.22.3 the proper functioning and operation of User Terminal Devices and the testing of the Multimedia Service Network or the testing of Multimedia Services in relation thereto;
- 1.2.23 "Terminal Connection Point" means any point within an item of Terminal Connection Equipment at which Multimedia Traffic is conveyed to or from one or more items of User Terminal Devices;
- 1.2.24 "Territory" means the Republic of the South Africa;
- 1.2.25 "User Terminal Device" means any fixed or mobile device which does not form part of the Multimedia Service Network and which is used for the delivery of

Multimedia Services directly to customers, and has been type approved in accordance with Chapter VI of the Telecommunications Act;

- 1.3 Words and expressions defined in any clause shall, for the purposes of that clause, bear the meaning assigned to such words and expressions in such clause.
- 1.4 If any provision is a substantive provision conferring rights or imposing obligations on the Licensee, notwithstanding that it is only in the interpretation clause, effect will be given to it as if it were a substantive provision in the body of the Licence.
- 1.5 All terms and expressions not expressly defined herein shall have the meanings assigned to them as set out in the Telecommunications Act, as amended, and all applicable Regulations issued in terms of that Act. The Licensee is required to comply with the Telecommunications Act and all applicable Regulations.

2 LICENCE PERIOD

The initial term of the Licence shall be for a period of 15 (fifteen) years from the Effective Date.

3 LICENCE AUTHORITY

- 3.1 Subject to the provisions of section 32C(8) of the Telecommunications Act, the Licensee is authorised by this Licence to own, lease, construct, maintain, operate and use a Multimedia Service Network for the provision of Multimedia Services within the Republic.
- 3.2 The Licensee shall be entitled and is authorised in terms of this Licence to:
- 3.2.1 provide Multimedia Services to any person who requests such services and to use the same for the Licensee's business activities and own purposes;
- 3.2.2 provide the Multimedia Service as a common carrier on a reasonable, equitable and non-discriminatory basis;
- 3.2.3 use the frequency spectrum assigned by the Authority in the frequency spectrum licence for the provision of Multimedia Services;
- 3.2.4 connect or, cause or allow to be connected to the Multimedia Service Network, any telecommunication system in the Territory which is licensed under the Telecommunications Act or which the Authority has prescribed can

- be provided without a licence in terms of section 33 to the Telecommunications Act;
- 3.2.5 connect or, cause or allow to be connected to the Licensee's Multimedia Service Network, any telecommunication system or telecommunication service situated outside the Territory except where such connection would contravene a treaty or other agreement to which the Territory is a party;
- 3.2.6 connect or, cause or allow to be connected to the Multimedia Service Network any User Terminal Device;
- 3.2.7 connect or, cause or allow to be connected to the Multimedia Service Network any telecommunication facility required for the provision of Multimedia Services;
- 3.3 In addition to the rights referred to in clause 3.2, the Licensee shall be entitled to:
- 3.3.1 procure, design, develop and maintain any software or hardware to be used in connection with any telecommunication system and to use for the provision of its Multimedia Service;
- 3.3.2 use any international protocol application in its provision of the Multimedia Service;
- 3.3.3 use its own telecommunication facilities in the provision of the Multimedia Service and the interconnection of the same to any telecommunication system;
- 3.3.4 provide any service which is ancillary to the Licensee's provision of a Multimedia Service, including billing and installation, maintenance and repair services; and
- 3.3.5 enter into suitable arrangements with any Operator for the sharing of infrastructure and facilities, for the co-location of facilities and infrastructure and to interconnect with any Operator.
- 3.4 All the rights set out in this Licence shall be subject to the provisions of the Telecommunications Act, the relevant Regulations promulgated from time to time in terms of section 96 and the terms and conditions of this Licence.
- 3.5 The Licensee shall be entitled by virtue of this Licence to operate the Multimedia Service Network and to provide Multimedia Services together with all or any other rights granted to the Licensee under this Licence.

- 3.6 Nothing contained in this Licence shall authorise the Licensee to provide telecommunication facilities to anyone licensed under the Telecommunications Act to provide value-added network services, in circumstances where that person, by obtaining the telecommunication facilities in question from the Licensee, violates a term of its own telecommunication licence or otherwise acts in a manner inconsistent with that contemplated by section 40(2) of the Telecommunications Act. {Sentech requests that this clause be deleted in its entirety}
- 3.7 The Licensee shall use its reasonable endeavours to ensure that its employees, agents or contractors comply with the provisions of this Licence in the exercise or performance of the Licensee's rights and obligations as set out in this Licence.

4 LICENCE FEES

4.1 Fixed Licence Fee

In consideration for the grant of the Licence, the Licensee shall pay to the Authority a fixed licence fee in respect of the initial term of the Licence, referred to in clause 2, in the amount of R25m (twenty five) million Rand.

AND THE PROPERTY.

- 4.1.1 Should the Licensee fail to make payments of any amounts due in terms of clause 4.1.2 on the due date for payment thereof, then and in such event, interest at the Applicable Interest Rate shall accrue on the amount which is due in terms of clause 4.1.2, such interest to be calculated from the due date for payment thereof to the date of actual payment.
- 4.1.2 Subject to clause 4.1.4, the principal amount of the fixed licence fee referred to in clause 4.1 shall be payable by the Licensee to the Authority in 5 (five) equal annual instalments, the first instalment to be paid on the third anniversary of the Effective Date.
- 4.1.3 The Licensee may accelerate payments of the fixed licence fee referred to in clause 4.1. by making payment prior to the dates specified in 4.1.2 and, in such cases, the provisions of clause 4.1.2 shall still apply.
- 4.1.4 In addition to the fixed licence fee, the Licensee shall pay the fees stipulated in its frequency spectrum licence and such other fees as may be payable in terms of such Regulations as may apply to it from time to time.

4.2 Annual Variable Licence Fee

- 4.2.1 Subject to what is stated below, an annual variable licence fee in an amount equal to 0.5% (one half of one percent) of the audited Licence Fee Income shall be payable by the Licensee to the Authority.
- 4.2.2 The first payment of the annual variable licence fee referred to in clause 4.2.1 shall be made within 3 (three) months of the Licensee's Financial Year in the third year of the Licence Period and shall be calculated on the basis of the Licence Fee Income generated during the third year of the Licence Period.
- 4.2.3 After the fourth year of the Licence Period, the annual variable licence fee shall be payable within ninety days of the expiry of Sentech's financial year in question.
- 4.2.4 The annual variable licence fee payable by the Licensee shall be reviewable by the Authority after the fifth anniversary of the Effective Date, provided that, in the event of the percentage of Licence Fee Income payable by the Licensee being increased by the Authority, the percentage so increased shall not exceed the percentage of net operating income payable by similarly licensed operators in respect of the corresponding licence fees payable by them.
- 4.2.5 In reviewing the percentage of Licence Fee Income payable by the Licensee in terms of this clause, the Authority shall have due regard to factors such as the number of similarly licensed operators in the Territory, the spectrum allocated to such similarly licensed operators and the percentage of net operating income payable by such similarly licensed operators.
- 4.2.6 The provisions of clause 4.2 shall apply *mutatis mutandis* to all subsequent Licence Periods of this Licence.

5 UNIVERSAL SERVICE FUND AND COMMUNITY SERVICE OBLIGATIONS

5.1 Universal Service Fund Obligations

The Licensee shall pay an annual contribution to the Universal Service Fund as prescribed by regulation in terms of section 67(2) of the Telecommunication Act.

5.2 Community Service Obligations

The Licensee shall establish and maintain School Internet Laboratories (computer laboratories with Internet Connectivity) in 500 schools in rural areas over 5 (five) years as specified in Annexure A. {Sentech requests that this clause be deleted in its entirety}

PERFORMANCE SPECIFICATIONS

- 6.1 The Multimedia Services provided by the Licensee shall conform to such Regulations as prescribed by the Authority.
- 6.2 The Licensee shall introduce measures and at all times use its resources to ensure that any telecommunication apparatus used in connection with the Multimedia Service Network or in the provision of Multimedia Services, including any telecommunication facilities owned by the Licensee shall comply with:-
- 6.2.1 the requirements for electromagnetic compatibility; and
- 6.2.2 the requirements for network interoperability, as prescribed by the Authority in accordance with international specifications and standards.
- 6.3 If requested by a customer in relation to a particular invoice, the Licensee shall provide such customer with a detailed statement of account.

7 CUSTOMER SERVICE STANDARDS

- 7.1 The Licensee shall develop and enforce guidelines for use by its personnel when handling enquiries and complaints from a customer to whom it provides Multimedia Services and to whom it supplies Terminal Connection Equipment and/or User Terminal Devices. The Licensee shall make such guidelines available to any customer who requests to be furnished with a copy of the same.
- 7.2 The Licensee shall prepare and enforce standard terms and conditions for the use of its Multimedia Services and it shall make such standard terms and conditions available to customers at the commencement of or the bringing into service of Multimedia Services to be provided to such customers.

8 FAIR TRADING

8.1 The Licensee, its agents and sub-contractors shall not demonstrate any undue preference to or exercise any undue discrimination against any person or class

- or description of persons in respect of the provision of Multimedia Services or in respect of the construction or maintenance of any connection to the Multimedia Service Network.
- 8.2 The Licensee shall include a provision in its end-user agreements with customers wherein it undertakes to comply with a code of practice for consumer affairs to be established by the Licensee and which shall at a minimum address the following:-
- 8.2.1 guidance to customers in respect of disputes or complaints relating to the provision of Multimedia Services by the Licensee;
- 8.2.2 advice to customers on charging and billing and procedures including the procedures to be followed in respect of billing and charging enquiries;
- 8.2.3 advice on the procedures to be followed in respect of the proper use of the Multimedia Services by customers; and
- 8.2.4 standards of conduct to be adhered to by the Licensee in the provision of Multimedia Services to its customers.
- 8.2.5 The Licensee shall lodge with the Authority representative sample copies of all end-user contracts relating to the Licensee's provision of Multimedia Services to any customer.
- 8.3 The Licensee shall not unfairly cross-subsidise its charges for any Multimedia Services in which the Licensee competes with the services mentioned in section 32C (7) of the Act.

9 ACCOUNTS

- 9.1 The Licensee shall maintain proper accounting records in a form, which is sufficient to show and explain its transactions and in this regard, the Licensee shall maintain such records as fairly represent the cost, revenue and financial position of the Licensee's business activities in terms of this Licence.
- 9.2 Within 3 (three) months after the end of each Financial Year of the Licensee, the Licensee shall present to the Authority audited annual financial statements prepared in accordance with Generally Accepted Accounting Practice.

10 TRANSFER OF LICENCE AND OWNERSHIP AND CONTROL

10.1 Transfer of Licence

The Licensee shall not transfer the Licence without the prior written consent of the Authority having been obtained in terms of the Telecommunications Act and the applicable Regulations.

10.2 Transfer of ownership and control

- 10.2.1 Any transfer of shares of the issued share capital of the Licensee being acquired by any party which is not a shareholder of the Licensee as at the Effective Date shall require prior written approval by the Authority.
- Any transaction, agreement or undertaking entered into by the Licensee with any other party that could have the effect of changing the control of the Licensee, shall require prior written approval by the Authority.

11 EMPLOYMENT EQUITY, HUMAN RESOURCE DEVELOPMENT AND TRAINING

- 11.1 The Licensee shall comply with the provisions of the Employment Equity Act, No 55 of 1998, and the Skills Development Act, No 97 of 1998, in relation to employment equity, human resource development and training. All reports submitted to the Department of Labour in this regard by the Licensee shall also be submitted to the Authority.
- 11.2 The Licensee shall participate in the development of the telecommunication industry by, inter alia:
- 11.2.1 supporting independent contractors from Historically Disadvantaged Groups; and
- 11.2.2 supporting industry development initiatives.

12 GENERAL

- 12.1 Should any provision of this Licence be invalid and unenforceable for any reason, the remaining provisions shall nevertheless remain of full force and effect.
- 12.2 To the extent that this Licence may be inconsistent with any provisions of the Telecommunications Act or any applicable Regulations, such provisions of the

Telecommunications Act or the Regulations shall take precedence over this licence.

13 FORCE MAJEURE

- 13.1 The Licensee shall not be liable for any default or delay in the performance of its obligations under the Licence if and to the extent:-
- 13.1.1 such default or delay is caused, directly or indirectly, by fire, flood, earthquake, elements of nature or acts of God, riots, civil disorders, rebellions or revolutions in any country or any other cause beyond the reasonable control of the Licensee; and
- 13.1.2 the Licensee is without fault in causing such default or delay.

14 CONFIDENTIALITY OF INFORMATION

14.1 Content of communications

The Licensee shall not disclose the content of any communication transmitted or intended for transmission over the Multimedia Service Network, unless required to do so by a court order or in terms of any law.

14.2 Customer information

The Licensee shall not use any information regarding its past, current or potential customers for purposes other than those for which the information was obtained, unless the customer gives prior written consent to such other use.

15 RENEWAL OF THE LICENCE

The Licence may be renewed in terms of the Telecommunications Act.

16 CANCELLATION OF LICENCE

In the event of the liquidation of the Licensee, the Licensee shall notify the Authority in writing of the cancellation of the Licence.

17 REVOCATION OF THE LICENCE

WITH CONTROL TO THE WAR IN THE

BALING LINEAU COLLEGE AND AND AND FOR THE BE

The Licence may be revoked in accordance with the Telecommunications Act.

18 NOTICES

18.1 The Licensee selects as its domicilia citandi et executandi the following physical addresses:-

Name	Physical Address	Telefax		
	*****	53.8		
Sentech Limited	Sender Technology Park	(011) 471-4753		
	Octave Road	16 J S 30		
n to	Radiokop			

or such other address or telefax number as may be substituted by notice given as herein required. The Licensee shall be entitled from time to time by written notice to the Authority, to vary its *domicilium* to any other address within the Territory.

- 18.2 Any notice addressed to the Licensee at its physical address shall be delivered by hand, or sent by telefax.
- 18.3 Any notice shall be deemed to have been given:-
- 18.3.1 if hand delivered during business hours on a business day, on the day of delivery;
- 18.3.2 if sent by telefax, on the date of sending of such telefax.

19 AMENDMENT

- 19.1 Subject to the provisions of section 48 to the Telecommunications Act, the Authority shall be entitled to amend any term or condition of this Licence.
- 19.2 No amendment, alteration, addition, variation or consensual cancellation of this Licence shall be of any force or effect unless reduced to writing and signed by the Authority.

22. MISCELLANEOUS PROVISIONS

- 22.1 Adherence to International Standards
- The Licensee shall comply with the applicable standards and requirements of the ITU, as agreed to or adopted by the Republic.

22.2 Compliance

The Licensee shall comply with all applicable laws of the Republic and with all international obligations to which the Republic is bound. Unless specifically indicated to the contrary, nothing in this Licence shall be construed or understood

so as to relieve or exempt the Licensee from complying with the provisions of this Licence or the Telecommunications Act. The Licensee shall use all reasonable endeavours to ensure that its employees, agents or contractors comply with the provisions of this Licence and the Telecommunications Act.

22.3 Access to Network Facilities

Upon reasonable notice, the Licensee shall provide access to its Network facilities for inspection by the Authority.

ssued at	 on this	(day)	<i>I</i>	_(mon	th) / 2002
				e e e e e e e e e e e e e e e e e e e	
		IDLA LANGA AIRPERSON	+		D=-

NKATEKO NYOKA

CHIEF EXECUTIVE OFFICER
INDEPENDENT COMMUNICATIONS AUTHORITY OF SOUTH AFRICA (ICASA)

Sentech requests that this annexure be deleted in its entirety ANNEXURE A

- A1 The Licensee shall provide the following for each School Internet laboratory:
 - A1.1 A certain number of workstations, dependent on the needs and size of the school, with:
 - Type A: Small rural schools with 5-computer Local Area Network with an Internet connection.
 - Type B: Medium rural schools with 15-computer Local Area Network with an Internet connection.
 - Type C: Large rural schools with 25-computer Local Area Network with an Internet connection.

In each year of the rollout of these licence conditions, 15% of schools should be Type A, 50% Type B and 35% Type C as appropriate to the needs of the schools.

The Authority may, upon application, vary the above-mentioned percentages.

- A1.2 Workstations with the following functionality:
 - A1.2.1 Password-protected logins for each user in the school (up to 1,500 potential users):
 - A1.2.2 Internet access from each workstation;
 - A1.2.3 Network-based store-and-forward email system, with email client on each workstation;
 - A1.2.4 Ability to run common office applications (including word processor, spreadsheet, and presentation software);
 - A1.2.5 Shared printing:
 - A1.2.6 Ability to save files on server:
 - A1.2.7 Anti-virus software;
 - A1.2.8 Backup facilities ability to copy data to offline backup media; and
 - A1.2.9 At least a third of the workstations in each laboratory should be able to run current educational multimedia applications, with sound and video functionality.
- A1.3 Internet Connectivity with sufficient bandwidth to allow internet usage from all workstations is to be provided by VSAT or other appropriate technology.

 Reasonable monthly connectivity is to be provided to the school, at a rate agreed in the project implementation planning process stipulated in A3.

- A1.4 Appropriate furniture for an Internet Laboratory.
- A1.5 Refurbishment and upgrade to buildings where appropriate to provide a secure and useable Computer Laboratory.
- A1.6. Technical training to at least 2 teachers from each school to act as network administrators, to enable them to perform common network maintenance functions, including adding and removing users and email addresses.
- A1.7. Professional development opportunities to at least 5 teachers from each school, to enable them to use the ICT facilities in support of more effective teaching and learning.
- A2 The Licensee shall establish and maintain computer laboratories with Internet Connectivity in schools in rural areas according to the following implementation schedule:
 - A2.1 Within 2 (two) years of the Effective Date, the Licensee shall have installed a total of 100 School Internet Laboratories in rural areas.
 - A2.2 Within 3 (three) years of the Effective Date, the Licensee shall have installed a total of 200 School Internet Laboratories in rural areas.
 - A2.3 Within 4 (four) years of the Effective Date, the Licensee shall have installed a total of 300 School Internet Laboratories in rural areas.
 - A2.4 Within 5 (five) years of the Effective Date, the Licensee shall have installed a total of 400 School Internet Laboratories in rural areas.
 - A2.5 Within 6 (six) years of the Effective Date, the Licensee shall have installed a total of 500 School Internet Laboratories in rural areas.
 - The proportions of School Internet Laboratories with 5, 15 and 25 workstations will be as set out in A1.1.
 - A2.6 During the sixth year of the Licence, the Authority may review the community service obligations in order to determine the Licensee's obligations for the remainder of the Licence period.
- A3. The licensee has a period of 6 months following the Effective Date to plan for the implementation of these School Internet Laboratories.

- A3.1 In this time, consultations must be carried out with national and provincial Departments of Education, and other appropriate bodies such Schoolnet SA.
- A3.2 An implementation plan must be produced by the end of this 6 month period, which will stipulate the technology to be used, operational details, cost-sharing between the various parties involved and specify by name the schools to which School Internet Laboratories will be rolled out in each year.
- A3.3 An implementing body is to be established to manage the process of rollout of the School Internet Laboratories, funded by the Licensee. This body will liase with the Licensee, the Authority, the Departments of Education and other relevant parties and manage the rollout. The responsibilities will include management of the preparation of sites, delivery of equipment, training and professional development, support, maintenance and ensuring monitoring and evaluation of the process. This should be housed in an independent organisation with experience in delivery of information and Communication Technology (ICT) projects to schools.
- A3.4 Should the Licensee fall to produce an acceptable implementation plan which is agreed by the Licensee, the Authority and the national and provincial Department of Education within this 6 month period, the Authority may impose an implementation plan upon the Licensee.
- A4. For the purpose of determining whether the Licensee has complied with its obligations as set out in A2, the Authority and the Licensee shall use their best endeavours to reach agreement, within 6 months of the Effective Date, on the applicable performance indicators. Should the Authority and the Licensee fall to agree with this six month period, the Authority may impose a set of performance indicators. The implementing body specified in A3.3 shall outsource the monitoring of these obligations, and report to the Authority on a quarterly basis.
- A5. The Implementation of the Licensee's obligations may, subject to the provisions hereof, be coordinated between the Licensee, similar licensees, national and provincial Departments of Education and others working to achieve computer and internet accessibility in rural schools.
- A6. Subsequent to the second anniversary of the Effective Date, the Authority shall periodically assess the Licensee's level of compliance with its obligations to establish and maintain School Internet Laboratories in rural areas in accordance with the implementation timetable set out in A2.

A7. In the event of the Licensee failing to comply with any of its abovementioned obligations, the Authority shall be entitled, after due investigation, to impose penalties in terms of section 100 of the Telecommunications Act.

a pay a hepadeyar an ex god to d

and the second of the second

The world property of the second

Tarker of the and replaced in 1991, the area of the second

and the profile and a state of the same and a sign of the same and the

ည္းကို အရည္ရွိနဲ့ မီးအလည္ကို ရွိႏွံုး အေလ နည္ လူသူ ရသည္ေတြ ေတြေသတ္ေတြကို သည္။ ကို သည္ အလုိ ပြဲခ်စ္သည့္သို့ မွာအသ

and the comment to be the particle of the same in the

MANDLA LANGA CHAIRPERSON ICASA

Dog ate your Gazette? ... read it online



www.SA Gazettes.co.za

A new information Portal keeping you up to date with news, legislation, the Parliamentary programme and which is the largest pool of SA Gazette information available on the Web.

- · Easily accessible through the www!
 - Government Gazettes from January 1994
 - Compilations of all Indexes pertaining to the past week's Government Gazettes
 - All Provincial Gazettes from September 1995
 - Parliamentary Bills as of January 1999
- · Available in full-text, with keyword searching
- Sabinet Online scans, formats, edits and organize information for you. Diagrams and forms included as images.
- No stacks of printed gazettes all on computer. Think of the storage space you save.
- Offers Bill Tracker complementing the SA Gazettes products.

For easy electronic access to full-text gazette info, subscribe to the SA Gazettes from Sabinet Online. Please visit us at www.sagazettes.co.za





Looking for back copies and out of print issues of the Government Gazette and Provincial Gazettes?

The National Library of SA has them!

Let us make your day with the information you need ...

National Library of SA, Pretoria Division PO Box 397 0001 PRETORIA

Tel.:(012) 321-8931, Fax: (012) 325-5984

E-mail: infodesk@nlsa.ac.za



Soek u ou kopieë en uit druk uitgawes van die Staatskoerant en Provinsiale Koerante?

Die Nasionale Biblioteek van SA het hulle!

Met ons hoef u nie te sukkel om inligting te bekom nie ...

Nasionale Biblioteek van SA, Pretoria Divisie Posbus 397

0001 PRETORIA

Tel.:(012) 321-8931, Faks: (012) 325-5984

E-pos: infodesk@nlsa.ac.za

Printed by and obtainable from the Government Printer, Bosman Street, Private Bag X85, Pretoria, 0001

Publications: Tel: (012) 334-4508, 334-4509, 334-4510
Advertisements: Tel: (012) 334-4673, 334-4674, 334-4504
Subscriptions: Tel: (012) 334-4735, 334-4736, 334-4737
Cape Town Branch: Tel: (021) 465-7531

Gedruk deur en verkrygbaar by die Staatsdrukker, Bosmanstraat, Privaatsak X85, Pretoria, 0001

Publikasies: Tel: (012) 334-4508, 334-4509, 334-4510 Advertensies: Tel: (012) 334-4673, 334-4674, 334-4504 Subskripsies: Tel: (012) 334-4735, 334-4736, 334-4737

Kaapstad-tak: Tel: (021) 465-7531